Author's response to reviews

Title: A quantitative content analysis of UK newsprint coverage of children's exposure to second-hand smoke and legislation targeting smoking in vehicles carrying children

Authors:

Chris Patterson (chris.patterson@glasgow.ac.uk)
Sean Semple (sean.semple@abdn.ac.uk)
Karen Wood (karen.wood@glasgow.ac.uk)
Sheila Duffy (SDuffy@ashscotland.org.uk)
Shona Hilton (shona.hilton@glasgow.ac.uk)

Version: 2 Date: 14 January 2015

Author's response to reviews:

Dear Editor,

We have resubmitted our manuscript with the requested revisions, as detailed below:

1) Added line and page numbers
2) Added a Conclusions section
3) Added a statement of ethical approval to the end of the Methods section
4) Moved tables to after the references section

We hope that these changes are appropriate and adequate.

Regards,

Chris Patterson